ORIGINAL RESEARCH

Patient-Reported Burden of Illness and Unmet Needs in *Demodex* blepharitis

Preeya K Gupta^[b], Vance Thompson², Leslie O'Dell², Anh N Ho³, Arthur Chan³, Bhagyashree Oak⁴, Amod Athavale⁴, Elizabeth Yeu^[b]

¹Triangle Eye Consultants, Raleigh, NC, USA; ²Vance Thompson Vision, Sioux Falls, SD, USA; ³Tarsus Pharmaceuticals, Inc, Irvine, CA, USA; ⁴Trinity Life Sciences, Waltham, MA, USA

Correspondence: Amod Athavale, Trinity Life Sciences, 230 3rd Ave, Waltham, MA, 02451, USA, Email aathavale@trinitylifesciences.com

Background: *Demodex* blepharitis is an ocular disorder caused by the infestation of *Demodex* mites that reside on the eyelash follicles.

Objective: This study assessed the clinical, humanistic, and economic burden of *Demodex* blepharitis from a patient perspective.

Methods: This cross-sectional, observational study used a web-enabled survey to collect data from US adults with *Demodex* blepharitis in 2022. Patients with unique burdens, including those receiving dry eye disease medications, wearing contact lenses, or experiencing cataracts or glaucoma were also examined.

Results: Among 113 patients, mean age was 48.5 years (standard deviation $[SD] \pm 13.6$). Half had private/commercial insurance, and 55% had Medicare and/or Medicaid. Patients had *Demodex* blepharitis for an average of 4.3 years (SD \pm 6.7 years) before the study, and 1.2 years (SD \pm 3.0 years) elapsed between the appearance of symptoms and diagnosis. Common symptoms, including redness, dryness, itchiness of the eyelids, and itchiness of the eyes, persisted or returned shortly after diagnosis and disease management in most patients, and they were associated with a negative impact on quality of life. Patients visited their healthcare practitioner for *Demodex* blepharitis a mean of 3.9 times (SD \pm 4.8) in the preceding year. Patients were often managed with off-label prescription medications, such as medications indicated for dry eye disease, in-office procedures, or over-the-counter management options.

Discussion: Patients with *Demodex* blepharitis reported symptoms impacting their quality of life and activities of daily living, which persisted after diagnosis and disease management. This suggested that the effectiveness of the reported symptom management options was temporary and highlighted an unmet need in treating the root cause of the disease.

Conclusion: Patients with *Demodex* blepharitis were symptomatic, and the commonly used management options for *Demodex* blepharitis lacked long-term symptom relief or mite eradication, demonstrating a high unmet need in treating patients with *Demodex* blepharitis.

Plain Language Summary: *Demodex* blepharitis, a common chronic eyelid disease, was surveyed among US adults. Patients reported several doctor visits for diagnosis. After disease management therapies, patients reported symptoms continued or returned quickly. Unresolved symptoms negatively affected patients' lives, and symptomatic therapy management was a long-term expense that did not provide long-term relief. Patients frequently visited doctors due to recurring symptoms, highlighting a high unmet need in treating patients with *Demodex* blepharitis.

Keywords: Demodex blepharitis, burden of illness, disease management, quality of life

Introduction

Blepharitis is a common ocular disorder characterized by inflammation of the eyelid margin, redness, and ocular irritation.^{1–3} One of the most common causes of blepharitis is infestation with *Demodex* mites that reside on the eyelash follicles and meibomian glands.^{4,5} *Demodex* mites are the most common human ectoparasites, and at high population densities, *Demodex* infestations can cause damage to hair follicles and meibomian glands, disrupt the local microbiome,

© 2025 Gupta et al. This work is published and licensed by Dove Medical Press Limited. The full terms of this license are available at https://www.dovepress.com/terms work you hereby accept the Terms. Non-commercial uses of the work are permitted without any further permission from Dove Medical Press Limited, provided the work is properly attributed. For permission for commercial use of this work, please see paragraphs A2 and 5 of our Terms (https://www.dovepress.com/terms.php).

647

and induce delayed hypersensitivity responses.^{6,7} Globally, a wide range of prevalence estimates for *Demodex* blepharitis, from 41% to 70%, has been reported.^{8–13} A US-based real-world prevalence study recently revealed that 58% of patients visiting eye care clinics for any reason had *Demodex* blepharitis,⁵ suggesting that up to 25 million Americans may be affected by this condition.^{14,15} This prevalence among patients presenting for routine visits to ophthalmologists and optometrists also suggests that *Demodex* blepharitis is as common as dry eye disease (DED).⁵

Because an infestation by *Demodex* mites leads to the formation of pathognomonic lash-cuffing collarettes around the base of the eyelashes, *Demodex* blepharitis can be diagnosed if these collarettes are present.^{8,16–23} Collarettes can be accurately identified by having patients look down during an examination with a standard slit-lamp,^{6,17} (Figure 1) which is used by optometrists and ophthalmologists in routine eye exams.¹⁶ However, despite readily available diagnostic techniques, almost 50% of the patients with collarettes may not be diagnosed with *Demodex* blepharitis, suggesting substantial underdiagnosis resulting from misdiagnosis or missed diagnosis.⁵ Assessing patients for *Demodex* blepharitis is currently not part of the routine eye exam, leading to a general lack of awareness among eye care professionals about the importance of identifying collarettes in patients.⁵ *Demodex* blepharitis may also be underdiagnosed or misdiagnosed partly due to a substantial overlap of symptoms with DED and other ocular surface disorders.⁶

Typically, symptom management includes off-label prescription medications such as those indicated for DED, antibiotics, and anti-parasitics, alongside in-office lid cleansing measures such as microblepharoexfoliation, and overthe-counter (OTC) lid scrubs and wipes.⁶ The use of medications and therapies that do not directly address the mite infestation or lack proven efficacy leads to disease progression, chronic symptoms, and incurring long-term costs.²⁴

Chronic symptoms persist among *Demodex* blepharitis patients, with 51% experiencing them for 4 years or more.²⁵ Untreated, *Demodex* blepharitis has many ocular sequelae including lid and eyelash abnormalities such as trichiasis,



Figure I Illustration of Pathognomonic Lash-cuffing Collarettes of *Demodex* Mite Infestation as Identified by Examination with A Standard Slit-lamp. Notes: Lash-cuffing collarettes are visible at the base of the eyelashes of a patient with *Demodex* blepharitis while looking down during a standard slit-lamp examination. Image Credit: Vance Thompson, MD, Sioux Falls, SD, USA.

misdirected and thinning lashes, noticeable eye and eyelid erythema, corneal ectasia, and contact lens intolerance.^{2,4,26–31} Other findings include recurring hordeola and chalazia, meibomian gland disease and meibomian gland atrophy, progressive pterygium, and other corneal and conjunctival complications.^{29,32} In a US-based study involving patients with *Demodex* blepharitis across 20 eye care clinics, over 99% of participants reported experiencing at least one symptom.²⁴ Additionally, 77.4% stated that *Demodex* blepharitis significantly impacted their daily activities.²⁴

Efforts to quantify the clinical, humanistic, and economic burden of *Demodex* blepharitis remain scarce. This study sought to characterize the real-world, holistic burden of illness due to *Demodex* blepharitis from a patient perspective, including patient-reported outcomes before and after disease management, unmet need and challenges in diagnosis and current management, impacts on patient's quality of life, and healthcare resource utilization and costs.

Methods

Study Design

This cross-sectional, observational study used a web-enabled survey (developed with a third-party partner, MedSurvey) to collect anonymous information regarding the burden of *Demodex* blepharitis from qualified respondents. The study was conducted in accordance with the 2013 version of the Declaration of Helsinki, and applicable national and local requirements for good pharmacovigilance practices and complied with applicable respondent privacy regulations/ guidance as required by local law.³³ Before enrollment of respondents into the study, the protocol, screener, questionnaire, and other written information to be provided to respondents was reviewed and exempted by an independent Institutional Review Board (Advarra, Columbia, MD). All patients provided informed consent to participate in the survey. All survey respondents were identified using a unique identification code to maintain anonymity for study researchers. All data were de-identified, and no diagnostic or therapeutic interventions were used.

Recruitment

A multi-modal recruitment approach, using patient panels and eye-care provider (ECP)- referred recruits, was employed to recruit adult patients with *Demodex* blepharitis in the US. The study was conducted from June 6, 2022, until August 16, 2022, during which there were no approved treatments by the US Food and Drug Administration (FDA) for *Demodex* blepharitis. To maintain the anonymity of all patients, study coordinators, and study sponsors were masked through the recruitment process. Qualified respondents who provided informed written consent to participate were asked to complete a screening questionnaire based on the inclusion/exclusion criteria to determine if they met the study eligibility criteria.

Eligibility Criteria

Individuals were eligible for inclusion in this study if they were between 21 and 85 years of age. The lower age limit of 21 was selected to align with the age of majority in Mississippi, where it is 21 years, as compared to 18 years or 19 years in most US states. The upper limit of 85 years was chosen due to the increased likelihood of cognitive impairments in older individuals, which could affect their ability to provide informed consent or accurately complete surveys. Additional inclusion criteria required participants to be currently residing in the US, have a diagnosis of *Demodex* blepharitis provided by a doctor for at least 6 months prior to the survey (to ensure that patients had some treatment experience), can complete the survey in English, and have access to a smartphone and/or computer and internet connectivity to complete the web-enabled questionnaire.

Individuals were excluded from this study if they were directly affiliated with or had immediate family affiliated with an advertising agency in the healthcare area or a market research company in the healthcare area, the US FDA, a pharmaceutical or biotechnology firm, or a health economics and outcomes research company.

Survey Description and Data Collection

The web-enabled questionnaire was designed to collect information about the holistic burden of illness across multiple dimensions, including clinical, psychological, social, humanistic, and economic domains. Before fielding the survey, two telephone-based, 60-minute cognitive or pilot interviews were conducted with eligible respondents to assess

comprehension of the questions as intended; verify that the instrument was appropriately timed; and confirm that questions were appropriately sensitive to patient experiences.

The survey contained custom questions intended to capture patient demographic information; the diagnostic journey starting from the time symptoms prompted the first healthcare provider (HCP) visit; healthcare resource utilization; and therapies used for symptom management including off-label prescriptions, OTC therapies, and in-office management options. Several questions in the survey were answered using a 5-point scale. These included the impacts of symptoms on patient's quality of life and emotional states, for which the 5-point scale ranged from 1 (no impact at all) to 5 (very significant impact). The impact of symptoms experienced before diagnosis with *Demodex* blepharitis was assessed on a scale ranging from 1 (did not experience at all) to 5 (experienced significantly). Finally, the 5-point scale used to assess the willingness of patients to try new treatments for *Demodex* blepharitis ranged from 1 (completely disagree) to 5 (completely agree).

Validated Outcome Measures

The 12-item Ocular Surface Disease Index (OSDI) questionnaire was used to assess the severity of eye dryness and other ocular surface symptoms and their impact on vision-related functioning,³⁴ and the University of North Carolina Dry Eye Management Scale (UNC DEMS), a 1-item graded scale (1–10) was used to assess symptoms and their effect on patients' quality of life.³⁵ Impairments to work and daily activities were assessed using the 6-item Work Productivity and Activity Impairment (WPAI) questionnaire.^{36,37}

Data Analyses

Data were reported using descriptive statistics, including means and standard deviation (SD) for continuous variables, and medians with interquartile ranges (IQR) for ordinal data, such as those derived from 5-point agreement scales. Frequencies and percentages are provided for categorical variables. Analyses were conducted at the aggregate level, with no patient-level characteristics were included to protect patient identification. All statistical analyses were conducted in Q Research Software 5.12.4.0 (Glebe, Australia).

Results

Patient Demographic and Clinical Characteristics

The survey sample comprised 113 patients with an ECP-confirmed diagnosis of *Demodex* blepharitis for at least 6 months prior to the survey (Supplemental Table 1). Respondents had a mean age of 48.5 years (SD \pm 13.6) and included 50% with private/commercial insurance coverage, 55% with Medicaid or Medicare, 1% with other government insurance, and 2% who were uninsured. Patients reported living with clinically confirmed *Demodex* blepharitis for an average of 4.3 years (SD \pm 6.7) since their initial diagnosis.

Clinical Burden of Demodex blepharitis

Patients reported persistent symptoms over time, from before being diagnosed with *Demodex* blepharitis to the time of the diagnosis (Table 1). The responses revealed that 1% (1/113) of patients had no eye symptoms. In other words, nearly all (99%) of patients reported at least one symptom, and the most common symptoms included redness, dryness, itchiness of the eyelids, and itchiness of the eyes. At the time of diagnosis, patients reported symptoms occurring multiple times a day, including burning or stinging in eyes (32%; 7/22), dryness of the eyes throughout the day (24%; 10/42), and crusted eyes/eyelashes (24%; 4/17) (Supplemental Figure 1).

Despite the use of management options, patients reported that many symptoms from before *Demodex* blepharitis diagnosis recurred multiple times, specifically crusted eyes/eyelashes throughout the day, dryness of eyes upon waking in the morning, redness of the eyes, sensitivity to light, and eye inflammation (Figure 2). Further, patients reported the impact of their recurring eye symptoms on several outcomes using a 5-point scale of severity, from 1 (did not experience at all) to 5 (experienced significantly). Moderate-to-severe impacts (scores 3–5) were most frequently reported for unrefreshing sleep (74%; 84/113), irritation (71%; 80/113), fatigue (70%; 79/113), headaches (68%; 77/113), difficulty in falling asleep, staying asleep, or waking on time (67%; 76/113), and anxiety (65%; 74/113) (Supplemental Figure 2).

Symptom, n (%)	Prior to diagnosis N = 113	At diagnosis N = 113
Redness of the eyes	55 (49%)	55 (49%)
Dryness of the eyes throughout the day	54 (48%)	49 (43%)
ltchy eyes	49 (43%)	47 (42%)
Itchy eyelids	51 (45%)	43 (38%)
Eye inflammation	43 (38%)	42 (37%)
Swollen eyelids	40 (35%)	41 (36%)
Burning or stinging in the eyes	34 (30%)	40 (35%)
Crusted eyes/eyelashes upon waking up in the morning	40 (35%)	39 (35%)
Dryness of the eyes upon waking up in the morning	34 (30%)	37 (33%)
Watery eyes	35 (31%)	36 (32%)
Crusted eyes/eyelashes throughout the day	29 (26%)	34 (30%)
Sensitivity to light	35 (31%)	32 (28%)
Difficulty driving at night	25 (22%)	30 (27%)
Blurry vision	30 (27%)	26 (23%)
Flaking of the skin around the eyes	27 (24%)	20 (18%)
Presence of collarettes/mites but no other eye symptoms	I (1%)	I (I%)

Table I Percentage of Patients with Demodex Blepharitis Symptoms Over Time

Notes: This question was multi-select and respondents may have provided more than one response, as appropriate (eg, presence of more than one symptom prior or at diagnosis).

Challenges in Obtaining Demodex blepharitis Diagnosis

On average, patients visited an HCP 3.5 times (SD \pm 2.8) before receiving a confirmed diagnosis of *Demodex* blepharitis, and 37% of patients noted that the original HCP sent them to another specialist (Supplemental Figure 3). Of the overall study sample, 45% (51/113) visited an ophthalmologist and 42% (47/113) visited an optometrist before being diagnosed with *Demodex* blepharitis (Supplemental Table 2). Similarly, 44% of patients were ultimately diagnosed by an ophthalmologist and 40% by an optometrist. On average, the time between the appearance of symptoms and a *Demodex* blepharitis diagnosis was 1.2 years (SD \pm 3.0).

Disease Management in Demodex blepharitis

Patients used off-label prescription medications at higher rates before being diagnosed with *Demodex* blepharitis (Figure 3). Although the proportion of patients using off-label prescription medications declined after diagnosis, off-label medication use persisted in a subset of patients ranging from 5% (6/113) for varenicline nasal spray up to 16% (18/ 113) for cyclosporine eye drops. The use of OTC therapies before the diagnosis of *Demodex* blepharitis and at the time of the survey included warm eye compresses, lubricating eye drops or artificial tears, at-home eyelid wipes or washes, tea tree eyelid cleanser, tea tree oil, oral supplements, eye inserts, and hypochlorous acid cleansers (Supplemental Figure 4). In-office therapies and procedures were reported in <10% of patients to manage *Demodex* blepharitis, including microblepharoexfoliation (7% before diagnosis and 8% currently), intense pulsed light therapy (7% before diagnosis and 6% currently), heating therapies (6% before diagnosis and 9% currently), punctal plugs (6% before diagnosis and 4% currently), and in-office eyelid scrub (5% before diagnosis and 6% currently). Overall, 5% of patients reported never using any form of therapeutic management for their *Demodex* blepharitis symptoms.



Symptom did not go away Symptom went away for a short-period but reappeared Symptom went away and did not reappear I do not know

Figure 2 Results of Management of Eye Symptoms Prior to Demodex Blepharitis Diagnosis (N=113). Patients reported the results from symptom management therapies on specific symptoms before their diagnosis. For each symptom, patients reported if the symptom went away and did not reappear, went away for a short period but later reappeared, did not go away, or if the patient did not know. Patients also reported the number of times the symptom reappeared before diagnosis with Demodex blepharitis.



Figure 3 Off-label Prescription Medications used in the Management of *Demodex* Blepharitis Across Time (N=113). Patients reported receiving various prescription medications before receiving a confirmed diagnosis of *Demodex* blepharitis, or currently after receiving a confirmed diagnosis of *Demodex* blepharitis.

Notes: DB, Demodex blepharitis. This question was multi-select and respondents may have provided more than one response, as appropriate (eg, respondent was taking multiple medications at one time).

DED Therapies and Symptom Severity in Patients with Demodex blepharitis

Overall, patients with *Demodex* blepharitis reported using any type of prescription medication for DED at rates of 29% (33/113) before diagnosis and 21% (24/113) at the time of the survey. More than half of the patients currently taking prescription DED medications (54%; 13/24) did not have a diagnosis of DED, consequently more than 10% (13/113) of all patients in the study were currently taking an off-label prescription medication for DED at the time of the survey.

All patients were assessed with the OSDI questionnaire and UNC DEMS scale to assess the severity and impact of eye symptoms. Based on the OSDI, 62% of patients with *Demodex* blepharitis could be classified as having severe DED based on symptom severity, with 17% classified as moderate and 11% classified as mild DED. Scores on the UNC DEMS showed that 17% of patients reported their symptoms had a high impact on their quality of life (score: 8–10), 69% indicated a moderate impact (score: 4–7), and 14% indicated a low impact (score: 1–3).

Comorbidities with Demodex blepharitis

Patient subgroups had substantial differences from the overall patient population in some results regarding the clinical burden of *Demodex* blepharitis. Patients with cataracts (n = 22) had a longer average interval since their diagnosis with *Demodex* blepharitis, mean 6.3 years (SD \pm 8.1), than the overall population (4.3 years [SD \pm 6.7]). Likewise, those with glaucoma (n = 11) reported an average of 6.6 years (SD \pm 7.2) since being diagnosed with *Demodex* blepharitis. Patients also visited ophthalmologists prior to their diagnosis more often if they had cataracts (68%) or glaucoma (82%) than the overall study sample (45%; 51/113). A small number of patients in these subgroups also reported indirect as well as direct clinical burdens of *Demodex* blepharitis. Half (7/14) of patients with cataracts experienced delays in cataract treatment, and 25% (2/8) of those who required cataract surgery experienced post-surgery complications due to their *Demodex* blepharitis. Likewise, 67% (4/6) of patients with glaucoma felt that their *Demodex* blepharitis had resulted in a need to pursue surgery for their glaucoma.

Several subgroups had a longer interval from symptom appearance to diagnosis. For all patients, the average time was 1.2 years (SD \pm 3.0) and required 3.5 HCP visits (SD \pm 2.8). Patients with cataracts waited an average of 2.9 years from the appearance of their symptoms to their diagnosis with *Demodex* blepharitis, and those with glaucoma waited 3.5 years, while patients who wore contact lenses averaged 4.6 visits to an HCP before receiving a diagnosis.

Finally, patients with glaucoma reported receiving in-office therapies at a higher rate (54%; 6/11) than the overall patient population before the diagnosis of *Demodex* blepharitis (15%; 17/113).

Humanistic Burden of Demodex blepharitis: Patient Perspectives and Impact

Patients reported the extent to which *Demodex* blepharitis impeded activities of daily living during the 12 months preceding the survey using a 5-point scale of impact ranging from 1 (no impact at all) to 5 (very significant impact) (<u>Supplemental Figure 5</u>). Activities with a high proportion of patients reporting impacts of 3, 4, or 5 included reading (73%), driving a car (71%), basic housework (64%), ability to work (58%), and wearing glasses or contact lenses (58%). Patients also indicated that *Demodex* blepharitis was detrimental to their sense of confidence about their appearance and even happiness/life satisfaction, with 65% of patients or more reporting an impact of 3, 4, or 5 in all emotional domains queried by the survey (Supplemental Figure 6).

Patients who were working or students at the time of the survey (n = 63) were notably impacted by *Demodex* blepharitis in the previous 7 days as assessed by the WPAI questionnaire, with scores of 75% for overall work impairment, including both impairment and absenteeism, 63% for impairment while working, and 60% for absenteeism. Similarly, patients had a score of 63% for impairment of regular activities outside of work due to *Demodex* blepharitis.

Patients were interested in new disease management options, with most patients (83%) reporting a need for curative treatment, and 78% were willing to use new treatment options if they would improve their symptoms. Nearly as many (76%) expressed a willingness to switch their current treatment if something more effective became available (Supplemental Figure 7).

Economic Burden of Demodex blepharitis

Patients in this study visited their HCP due to *Demodex* blepharitis a mean of 3.9 times in the 12 months preceding the survey (SD \pm 4.8) and visited urgent care facilities an average of 1.4 times (SD \pm 2.2). Importantly, 52% of patients visited an urgent care facility related to their *Demodex* blepharitis in the year before the survey. In-office therapies were associated with the highest average annual out-of-pocket expenses for patients (\$368), followed by HCP visits (\$192), prescription medicines (\$175), and OTC medications or management options (\$115).

Discussion

This is the first study to examine the clinical, humanistic, and economic burden of *Demodex* blepharitis and the ability to manage this burden with management options during the time when there were no FDA-approved treatments available in the market for the condition. Almost all (99%) patients with *Demodex* blepharitis reported frequent and bothersome symptoms like blurry vision, crusted eyes, and dryness, persisting even post-diagnosis and disease management. Off-label use of DED medications was a part of reported disease management therapies with notable more than half (54%) of the patients with *Demodex* blepharitis taking DED medications did not have diagnosis of DED. Persistent symptoms had a negative impact on patients' daily life, ability to engage in daily activities, and psychosocial well-being and incurred additional costs to patients. *Demodex* blepharitis potentially complicates the diagnosis and management of comorbid conditions like cataracts and glaucoma, leading to prolonging patients' burden and increased healthcare resource utilization. As there was no standard of care for eradication of the *Demodex* mites at the time of this study, symptom alleviation was temporary, at best, illustrating an unmet need in treating *Demodex* blepharitis.

This study also identified challenges associated with obtaining a *Demodex* blepharitis diagnosis from patients' perspective. More than one-third (37%) of patients were referred to multiple specialists during their diagnosis journey, with an average of 1.2 years from first symptoms to a confirmed diagnosis of *Demodex* blepharitis. The evidence of multiple HCP visits before a definitive diagnosis and the use of off-label prescriptions, such as DED medications, suggests a potentially high percentage of misdiagnoses in these patients.

A major gap in care for *Demodex* blepharitis is the long-term use of off-label use of prescription drugs (ie, medications indicated for DED, or oral ivermeetin) and OTC options that lack supporting clinical trial data regarding efficacy and safety in treating patients with Demodex blepharitis. Off-label use of oral ivermectin unnecessarily subjects patients to unwanted adverse effects including hepatitis, serious Mazzotti reaction, and Steven-Johnson and Lyell disease. $^{38-40}$ Additionally, tea tree oil, or its active ingredient terpinen-4-ol, is a widely available OTC option that is commonly used in *Demodex* blepharitis; however the literature suggests there is uncertainty associated with its effectiveness, and well-documented unwanted adverse effects.^{32,41} Tea tree oil was studied in a randomized study involving 281 patients with *Demodex* blepharitis by Koo et al.⁴² While the study found 23.6% of patients achieved mite eradication after using tea tree oil for 1 month, patients had to comply with a relatively burdensome regimen of weekly in-office procedure of lid scrubs with highly concentrated 50% tea tree oil applied over 30 minutes following topical anesthesia, in addition to twice-daily at-home lid scrubs. Terpinen-4-ol has also been found to have deleterious effects on meibomian glands observed in vitro.⁴³ Additionally, in-office procedures such as intense pulsed light (IPL) were often used in the management of *Demodex* blepharitis.⁶ However, IPL therapy was utilized by only 7 patients (6%) at the time of the study. Recent studies have found that IPL treatments may reduce the average mite count when used for up to 12 weeks of therapy or used in adjunct with other treatments, but the high cost of IPL treatment may be a major hindrance to patient adoption.44-46

The economic burden of *Demodex* blepharitis is under-recognized. Payers were exposed to costs for DED medications and other prescription medications that did not treat *Demodex* blepharitis, as well as the increased healthcare resource utilization from repeated office visits for chronic symptom management therapies. Additionally, self-reported out-of-pocket costs from patients may not accurately reflect the full financial burden of the disease, particularly when factoring in insurance contributions. These self-reports, for both off-label prescription medications and OTC management options, might substantially underrepresent the actual costs incurred by both payers and patients for what amounts to ineffective or temporarily effective disease management.

Published estimates of the economic burden of *Demodex* blepharitis are unavailable; however, a 2011 study reported direct costs of treating DED of \$3.84 billion per year for symptom management, physician visits, and nutritional supplements. Indirect costs were estimated at \$55.4 billion yearly for productivity loss.⁴⁷ A substantial portion of this cost may be associated with *Demodex* blepharitis given the reported rates of misdiagnosis and off-label DED treatments prescribed to patients with *Demodex* blepharitis.

Together, these results suggest that underdiagnosis, misdiagnosis, and ineffective disease management options contribute to the overall high cost and high burden of illness in *Demodex* blepharitis. An observational, prospective study, by O'Dell et al reported frequent and bothersome symptoms experienced by patients and that 80% of patients with

Demodex blepharitis experienced a negative impact on their psychosocial well-being.²⁵ Trattler et al 2022 study found that more than half of the patients presenting for cataract surgery had *Demodex* blepharitis, suggesting *Demodex* blepharitis lead to worsening of eye health post cataract surgery.⁵ The negative effects of *Demodex* blepharitis on the outcomes of comorbid conditions aligns with the results in this study.

Study Limitations

A cross-sectional study design was used for this study, making interpretation of the results subject to several limitations. Convenience sampling was used for this survey and included patients who were willing and able to participate in an online study, possibly skewing the sample population. While many demographic characteristics were collected for this study (eg, age, race, etc)., patient's sex or gender was not collected. This could be a potential limitation of the study; however, *Demodex* blepharitis is commonly reported among both males and females in the U.S.⁶ Responses to the survey may have been subject to recall bias. Additionally, patient-reported clinic visit frequency, and out-of-pocket costs were collected across a mixed population of commercial, Medicare, Medicaid, and uninsured patients, which may not accurately reflect healthcare resource utilization and cost analysis at a national level. Finally, responses may have been subject to social desirability bias with an underreport of socially undesirable attitudes and overreport of socially desirable attitudes, such as infrequent use of prescribed treatments, which can lead to a distorted understanding of treatment compliance. Conversely, they may overreport behaviors that are viewed positively, such as frequent healthcare visits or diligent use of OTC therapies.

Conclusion

The burden of illness study of patients with *Demodex* blepharitis underscored a substantial burden across clinical, humanistic, and economic domains. Patients faced multiple challenges in obtaining *Demodex* blepharitis diagnosis, visited health care providers multiple times for recurring symptoms, and received management options that lacked long-term symptom relief or mite eradication, highlighting the high unmet need in diagnosing and treating patients with *Demodex* blepharitis.

Abbreviations

DB, *Demodex* Blepharitis; DED, Dry Eye Disease; ECP, Eye-Care Provider; FDA, Food and Drug Administration; HCP, Healthcare Provider; IQR, Interquartile Ranges; OSDI, 12-item Ocular Surface Disease Index; OTC, Over-the-Counter; SD, Standard Deviation; UNC DEMS, University of North Carolina Dry Eye Management Scale; US, United States; WPAI, Work Productivity and Activity Impairment.

Ethics Approval and Informed Consent

Ethical approval was granted by the Advarra Institutional Review Board (Columbia, MD; <u>https://www.advarra.com/</u>) as an exemption under 45 CFR 46.104(d)(2) prior to study initiation. The research team confirms that this study did not cause any physical or mental harm to participants and adhered to the principles outlined in the Declaration of Helsinki. Electronic consent was obtained from all participants through specific questions in the instrument screener.

Consent for Publication

Study participants consented to the publication of their data anonymously on an aggregate basis.

Acknowledgments

Erik MacLaren, PhD provided medical writing services supported by Trinity Life Sciences. Data analysis support was provided by Latashree Goswami, MPH from Trinity Life Sciences.

Author Contributions

All authors made a significant contribution to the work reported, whether that is in the conception, study design, execution, acquisition of data, analysis and interpretation, or in all these areas; took part in drafting, revising or critically

reviewing the article; gave final approval of the version to be published; have agreed on the journal to which the article has been submitted; and agree to be accountable for all aspects of the work.

Funding

This study was funded by Tarsus Pharmaceuticals, Inc. Tarsus Pharmaceuticals, Inc had a role in the study design; analysis and interpretation of data; writing of the report; and the decision to submit the report for publication.

Disclosure

Amod Athavale and Bhagyashree Oak are employees, with Amod Athavale also holding equity in Trinity Life Sciences and they received funding for conducting the study from Tarsus Pharmaceuticals, Inc. Anh Ho, and Leslie O'Dell are employees and shareholders of Tarsus Pharmaceuticals, Inc, funder of the study. Arthur Chan was previously an employee of Tarsus Pharmaceuticals, Inc at the time of this study. Elizabeth Yeu was a board member and a consultant of Tarsus Pharmaceuticals, Inc, funder of the study, at the time of this study, and is currently an employee of Tarsus Pharmaceuticals, Inc. Preeya K. Gupta, Vance Thompson, and Elizabeth Yeu disclose consulting fees from Tarsus Pharmaceuticals, Inc. Dr Gupta reports personal fees from Alcon, Bausch and Lomb, Sight Sciences, AbbVie, and Viatris, outside the submitted work. Dr Thompson reports personal fees for consultancy and stock options from AdOM, Eyedetec, Sofia Biologics, Stuart Therapeutics, Surface Pharmaceuticals, Tarsus Pharmaceuticals; consultant research fees from Alcon, BVI, Carl Zeiss Meditec, Johnson & Johnson, Nordic Pharma; consultant, research and stock from Bausch & Lomb and Rayner, outside the submitted work. The authors report no other conflicts of interest in this work.

References

- 1. Amescua G, Akpek EK, Farid M, et al. Blepharitis preferred practice pattern(R). *Ophthalmology*. 2019;126(1):P56–P93. doi:10.1016/j. ophtha.2018.10.019
- 2. Putnam CM. Diagnosis and management of blepharitis: an optometrist's perspective. *Clin Optom.* 2016;8:71–78. doi:10.2147/OPTO.S84795
- 3. Wolffsohn JS, Arita R, Chalmers R, et al. TFOS DEWS II diagnostic methodology report. Ocul Surf. 2017;15(3):539-574. doi:10.1016/j. jtos.2017.05.001
- 4. Fromstein SR, Harthan JS, Patel J, Opitz DL. Demodex blepharitis: clinical perspectives. *Clin Optom*. 2018;10:57–63. doi:10.2147/OPTO.S142708 5. Trattler W, Karpecki P, Rapoport Y, et al. The prevalence of Demodex blepharitis in US eye care clinic patients as determined by collarettes:
- a pathognomonic sign. *Clin Ophthalmol*. 2022;16:1153–1164. doi:10.2147/OPTH.S354692
- 6. Rhee MK, Yeu E, Barnett M, et al. Demodex blepharitis: a comprehensive review of the disease, current management, and emerging therapies. *Eye Contact Lens.* 2023;49(8):311–318. doi:10.1097/ICL.00000000001003
- 7. Nicholls SG, Oakley CL, Tan A, Vote BJ. Demodex species in human ocular disease: new clinicopathological aspects. *Int Ophthalmol*. 2017;37 (1):303–312. doi:10.1007/s10792-016-0249-9
- Kasetsuwan N, Kositphipat K, Busayarat M, et al. Prevalence of ocular demodicosis among patients at tertiary care center, Bangkok, Thailand. Int J Ophthalmol. 2017;10(1):122–127. doi:10.18240/ijo.2017.01.20
- 9. Lee SH, Chun YS, Kim JH, Kim ES, Kim JC. The relationship between demodex and ocular discomfort. *Invest Ophthalmol Vis Sci.* 2010;51 (6):2906–2911. doi:10.1167/iovs.09-4850
- Murphy O, O'Dwyer V, Lloyd-McKernan A. Ocular Demodex folliculorum: prevalence and associated symptoms in an Irish population. Int Ophthalmol. 2019;39(2):405–417. doi:10.1007/s10792-018-0826-1
- 11. Sedzikowska A, Oseka M, Grytner-Ziecina B. Ocular symptoms reported by patients infested with Demodex mites. Acta Parasitol. 2016;61 (4):808-814. doi:10.1515/ap-2016-0112
- 12. Sedzikowska A, Oseka M, Skopinski P. The impact of age, sex, blepharitis, rosacea and rheumatoid arthritis on Demodex mite infection. Arch Med Sci. 2018;14(2):353–356. doi:10.5114/aoms.2016.60663
- 13. Wesolowska M, Knysz B, Reich A, et al. Prevalence of Demodex spp. in eyelash follicles in different populations. Arch Med Sci. 2014;10 (2):319–324. doi:10.5114/aoms.2014.42585
- 14. Wilson FA, Stimpson JP, Wang Y. Inconsistencies exist in national estimates of eye care services utilization in the United States. J Ophthalmol. 2015;2015:435606. doi:10.1155/2015/435606
- 15. Farid M Demodex blepharitis: simple diagnosis, challenging treatment. digital supplement. cataract & refractive surgery today. Available from: https://crstoday.com/articles/demodex-blepharitis-simple-diagnosis-challenging-treatment/demodex-blepharitis-simple-diagnosis-challenging-treatment/. Accessed January 12, 2022.
- 16. Bitton E, Aumond S. Demodex and eye disease: a review. Clin Exp Optom. 2021;104(3):285-294. doi:10.1111/cxo.13123
- 17. Gao YY, Di Pascuale MA, Li W, et al. High prevalence of Demodex in eyelashes with cylindrical dandruff. *Invest Ophthalmol Vis Sci.* 2005;46 (9):3089–3094. doi:10.1167/iovs.05-0275
- 18. Lopez-Ponce D, Zuazo F, Cartes C, et al. High prevalence of Demodex spp. infestation among patients with posterior blepharitis: correlation with age and cylindrical dandruff [Alta prevalencia de infestacion por Demodex spp. en pacientes con blefaritis posterior: correlacion con edad y caspa cilindrica]. *Arch Soc Esp Oftalmol.* 2017;92(9):412–418. doi:10.1016/j.oftal.2017.01.001

- Luo X, Li J, Chen C, Tseng S, Liang L. Ocular demodicosis as a potential cause of ocular surface inflammation. *Cornea*. 2017;36 Suppl 1(Suppl 1): S9–S14. doi:10.1097/ICO.00000000001361
- 20. Rabensteiner DF, Aminfar H, Boldin I, et al. Demodex mite infestation and its associations with tear film and ocular surface parameters in patients with ocular discomfort. *Am J Ophthalmol.* 2019;204:7–12. doi:10.1016/j.ajo.2019.03.007
- Randon M, Liang H, El Hamdaoui M, et al. In vivo confocal microscopy as a novel and reliable tool for the diagnosis of Demodex eyelid infestation. Br J Ophthalmol. 2015;99(3):336–341. doi:10.1136/bjophthalmol-2014-305671
- 22. Zhang AC, Muntz A, Wang MTM, Craig JP, Downie LE. Ocular Demodex: a systematic review of the clinical literature. *Ophthalmic Physiol Opt*. 2020;40(4):389–432. doi:10.1111/opo.12691
- 23. Zhong J, Tan Y, Li S, et al. The prevalence of demodex folliculorum and demodex brevis in cylindrical dandruff patients. *J Ophthalmol.* 2019;2019:8949683. doi:10.1155/2019/8949683
- 24. Barnett M, Simmons B, Vollmer P, et al. The impact of Demodex blepharitis on patient symptoms and daily life. *Optom Vis Sci.* 2024;101 (3):151–156. doi:10.1097/opx.00000000002111
- O'Dell L, Dierker DS, Devries DK, et al. Psychosocial Impact of Demodex Blepharitis. Clin Ophthalmol. 2022;16:2979–2987. doi:10.2147/OPTH. S374530
- Biernat MM, Rusiecka-Ziolkowska J, Piatkowska E, Helemejko I, Biernat P, Gosciniak G. Occurrence of Demodex species in patients with blepharitis and in healthy individuals: a 10-year observational study. Jpn J Ophthalmol. 2018;62(6):628–633. doi:10.1007/s10384-018-0624-3
- Tarkowski W, Moneta-Wielgos J, Mlocicki D. Demodex sp. as a Potential Cause of the Abandonment of Soft Contact Lenses by Their Existing Users. *Biomed Res Int.* 2015;2015:259109. doi:10.1155/2015/259109
- Chiang CC, Lin CL, Tsai YY, Peng CL, Liao YT, Sung FC. Patients with blepharitis are at elevated risk of anxiety and depression. *PLoS One*. 2013;8(12):e83335. doi:10.1371/journal.pone.0083335
- Liu J, Sheha H, Tseng SC. Pathogenic role of Demodex mites in blepharitis. Curr Opin Allergy Clin Immunol. 2010;10(5):505–510. doi:10.1097/ ACI.0b013e32833df9f4
- 30. Pflugfelder SC, Solomon A, Stern ME. The diagnosis and management of dry eye: a twenty-five-year review. *Cornea*. 2000;19(5):644–649. doi:10.1097/00003226-200009000-00009
- Hung KH, Tan HY, Chen HC, Yeh LK. Clinical characteristics and topographic findings of corneal ectasia in patients with symptomatic Demodex blepharitis. *Taiwan J Ophthalmol.* 2021;11(2):146–155. doi:10.4103/tjo.tjo_45_20
- 32. Gao YY, Di Pascuale MA, Elizondo A, Tseng SC. Clinical treatment of ocular demodecosis by lid scrub with tea tree oil. *Cornea*. 2007;26 (2):136–143. doi:10.1097/01.ico.0000244870.62384.79
- World Medical A. World medical association declaration of Helsinki: ethical principles for medical research involving human subjects. JAMA. 2013;310(20):2191–2194. doi:10.1001/jama.2013.281053
- Schiffman RM, Christianson MD, Jacobsen G, Hirsch JD, Reis BL. Reliability and validity of the ocular surface disease index. Arch Ophthalmol. 2000;118(5):615–621. doi:10.1001/archopht.118.5.615
- Grubbs J, Huynh K, Tolleson-Rinehart S, et al. Instrument development of the UNC Dry Eye Management Scale. Cornea. 2014;33(11):1186–1192. doi:10.1097/ICO.000000000000243
- 36. Greco G, Pistilli M, Asbell PA, Maguire MG, Dry Eye A; Management Study Research G. Association of severity of dry eye disease with work productivity and activity impairment in the dry eye assessment and management study. *Ophthalmology*. 2021;128(6):850–856. doi:10.1016/j. ophtha.2020.10.015
- 37. Reilly MC, Zbrozek AS, Dukes EM. The validity and reproducibility of a work productivity and activity impairment instrument. *Pharmacoeconomics*. 1993;4(5):353-365. doi:10.2165/00019053-199304050-00006
- Holzchuh FG, Hida RY, Moscovici BK, et al. Clinical treatment of ocular Demodex folliculorum by systemic ivermectin. Am J Ophthalmol. 2011;151(6):1030–1034.e1. doi:10.1016/j.ajo.2010.11.024
- Navel V, Mulliez A, Benoist d'Azy C, et al. Efficacy of treatments for Demodex blepharitis: a systematic review and meta-analysis. Ocul Surf. 2019;17(4):655–669. doi:10.1016/j.jtos.2019.06.004
- 40. Pacqué M, Muñoz B, Greene BM, White AT, Dukuly Z, Taylor HR. Safety of and compliance with community-based ivermectin therapy. *Lancet*. 1990;335(8702):1377–1380. doi:10.1016/0140-6736(90)91253-7
- Ngo W, Jones L, Bitton E. Short-term comfort responses associated with the use of eyelid cleansing products to manage demodex folliculorum. *Eye Contact Lens.* 2018;44(Suppl 2):S87–S92. doi:10.1097/ICL.00000000000415
- 42. Koo H, Kim TH, Kim KW, Wee SW, Chun YS, Kim JC. Ocular surface discomfort and Demodex: effect of tea tree oil eyelid scrub in Demodex blepharitis. *J Korean Med Sci.* 2012;27(12):1574–1579. doi:10.3346/jkms.2012.27.12.1574
- 43. Chen D, Wang J, Sullivan DA, Kam WR, Liu Y. Effects of terpinen-4-ol on meibomian gland epithelial cells in vitro. *Cornea*. 2020;39 (12):1541–1546. doi:10.1097/ico.00000000002506
- 44. Huo Y, Mo Y, Wu Y, Fang F, Jin X. Therapeutic effect of intense pulsed light with optimal pulse technology on meibomian gland dysfunction with and without ocular Demodex infestation. *Ann Transl Med.* 2021;9(3):238. doi:10.21037/atm-20-1745
- 45. Zhang W, Cao X, Yang L, Duan Y, Zhang W. Analysis of treatment efficacy of intense pulsed light (M22) for meibomian gland dysfunction with demodex mites. *Clin Cosmet Invest Dermatol.* 2023;16:3743–3751. doi:10.2147/CCID.S435723
- 46. Zhang X, Song N, Gong L. Therapeutic effect of intense pulsed light on ocular demodicosis. Curr Eye Res. 2019;44(3):250–256. doi:10.1080/ 02713683.2018.1536217
- 47. Yu J, Asche CV, Fairchild CJ. The economic burden of dry eye disease in the United States: a decision tree analysis. *Cornea*. 2011;30(4):379–387. doi:10.1097/ICO.0b013e3181f7f363

Patient Preference and Adherence

Dovepress Taylor & Francis Group

Publish your work in this journal

Patient Preference and Adherence is an international, peer-reviewed, open access journal that focusing on the growing importance of patient preference and adherence throughout the therapeutic continuum. Patient satisfaction, acceptability, quality of life, compliance, persistence and their role in developing new therapeutic modalities and compounds to optimize clinical outcomes for existing disease states are major areas of interest for the journal. This journal has been accepted for indexing on PubMed Central. The manuscript management system is completely online and includes a very quick and fair peer-review system, which is all easy to use. Visit http://www.dovepress.com/testimonials.php to read real quotes from published authors.

Submit your manuscript here: https://www.dovepress.com/patient-preference-and-adherence-journal

658 🖪 🗙 in 🔼